Author's response to reviews

Title: Titanium vs Absorbable tacks Comparative Study (T.A.C.S.): a multicenter, non-inferiority prospective evaluation during laparoscopic repair of ventral and incisional hernia. Study protocol for randomized controlled trial

Authors:

- Gianfranco Silecchia (gianfranco.silecchia@uniroma1.it)
- Giuseppe Cavallaro (giuseppe.cavallaro@uniroma1.it)
- Luigi Raparelli (luigi.raparelli@hotmail.com)
- Stefano Olmi (stefano.olmi@gmail.com)
- Gianandrea Baldazzi (gbaldazzi@casacura.it)
- Fabio C Campanile (campanile@surgical.net)

Version: 4  Date: 22 May 2015

Author's response to reviews:

Dear Editor-in-Chief,

we made further revision of the manuscript as requested